Author's response to reviews

Title: Combining complementary data sources improves prevalence estimates validity of chronic medical conditions

Authors:

Ueli Zellweger (uzellweg@ifspm.uzh.ch)
Matthias Bopp (bopp@ifspm.uzh.ch)
Barbara M. Holzer (barbara.holzer@usz.ch)
Sima Djalali (sima.djalali@usz.ch)
Vladimir Kaplan (Vladimir.Kaplan@spital-muri.ch)

Version: 5 Date: 25 April 2014

Author's response to reviews:

As suggested by the editorial office, we revised the title to make it a single sentence.

Since our primary aim is not to present prevalence estimates for Switzerland but to demonstrate how the combination of several imperfect data sources may offer a substantial surplus-value when analysing prevalences of chronic conditions, we omitted the country in the title.